Author's response to reviews

Title: Etiology and outcomes of acute kidney injury in chinese children: a prospective multi-center investigation

Authors:

Yan Cao (yizhuwen@163.com)
Zhu-Wen Yi (cyzwcn@163.com)
Hui Zhang (zhaohuiang@163.com)
Xi-Qiang Dang (linauu@126.com)
Xiao-Chuan Wu (liiauang@126.com)
Ai-Wen Huang (uihang@126.com)

Version: 2 Date: 18 May 2013

Author's response to reviews: see over
Dear Editor:

We would like to submit this manuscript entitled “Etiology and outcomes of acute kidney injury in Chinese children: a prospective multi-center investigation” as a scientific contribution for publication in BMC Urology. This paper is new, neither the entire paper nor any part of its content has been published or has been accepted elsewhere. It is not being submitted to any other journal. We believe the paper may be of particular interest to the readers of your journal because this manuscript determined the clinical features, etiology, and outcomes of acute kidney injury (AKI) in Chinese children.

Correspondence and phone calls about the paper should be directed to Zhu-Wen Yi at the following address, phone and fax number and e-mail address:

Sincerely,

Zhu-Wen Yi,
Division of Pediatric Nephrology,
Children’s Medical Center, the Second Xiangya Hospital, Central South University, Changsha, Hunan 410011,
P.R.China
Tel: +86 731 85292170
Fax: +86 731 85533525
E-mail: cyzwcn@163.com